Author's response to reviews

Title: Basic geriatric assessment does not predict in-hospital mortality after PEG placement

Authors:

Christine Smoliner (christine.smoliner@hospital-borken.de)
Dorothee Volkert (dorothee.volkert@aging.med.uni-erlangen.de)
Anke Wittrich (anke.wittrich@bv-geriatrie.de)
Cornel C Sieber (Cornel.Sieber@klinikum-nuernberg.de)
Rainer Wirth (rainer.wirth@hospital-borken.de)

Version: 2 Date: 7 December 2011

Author's response to reviews:

Dear Sir or Madam,

please find our revised version of the manuscript uploaded.

As requested in your pre-review revision we changed following points:
1. We added one sentence regarding ethics approval, which is not needed in Germany for retrospective anonymized database analysis.
2. We completed the paragraph on the authors contributions.
3. We changed the contact author with Dr. Rainer Wirth being the new contact author.

We thank you for consideration of our manuscript.

Yours sincerely,

Christine Smoliner